Author’s response to reviews

Title: Doctor and practice characteristics associated with differences in patient evaluations of general practice

Authors:

Hanne N Heje (hh@alm.au.dk)
Peter Vedsted (p.vedsted@alm.au.dk)
Ineta Sokolowski (ineta.s@alm.au.dk)
Frede Olesen (fo@alm.au.dk)

Version: 8 Date: 26 March 2007

Dear Editor

We hereby submit the final version of the manuscript. We made the followin requested formatting changes:
You wrote: Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT.
We changed the heading of the abstract.
You wrote: Please remove the keywords.
We removed the keywords.
You wrote: Please change Material and method -> methods
We changed the heading to "Methods"
You wrote: Please condense references e.g. [14-15], [13,15,22] not [14][15].
We went through the manuscript and changed the reference-format where needed according to your request.
You wrote: ge Reference List -> References.
We changed the heading to "References"
You wrote: and legend - The table title should appear above the table. The legend (or key) should appear below the table. Please remove the table legends section and place them elsewhere.
We put the "Legend for tables 3-7" under the heading of each of the tables.
You wrote: 2 - Unfortunately due to the format of the on-line manuscript we cannot incorporate boxes 1 and 2. Please either include the text within the manuscript, or change the title to 'Table 8, 9'. Please remember to update any references to 'Boxes 1 and 2' within the text.
We changed the heading of the boxes to Tables 1 and 2 - and adjusted the other table-headings accordingly.
You wrote: Reference all of your tables within the text, currently tables 3-7 aren't referenced.
We referenced the final five tables as well.

We hope that the manuscript will now be considered ready for publication. If not - please don't hesitate to contact us.
Yours sincerely
on behalf of the authors
Dr. Hanne Heje